Author's response to reviews

Title: Acupuncture and physical exercise for affective symptoms and health-related quality of life in polycystic ovary syndrome: Secondary analysis from a randomized controlled trial

Authors:

Elisabet Stener-Victorin (elisabet.stener-victorin@neuro.gu.se)
Göran Holm (goran.l.holm@vgregion.se)
Per Olof Janson (per-olof.janson@obgyn.gu.se)
Deborah Gustafson (deborah.gustafson@neuro.gu.se)
Margda Waern (margda.waern@neuro.gu.se)

Version: 2 Date: 25 May 2013

Author's response to reviews: see over